

# GOPEN ACCESS

**Citation:** Gabani MH, Ahmed AA, Hassan AA, Abdalla MA, Mustafa SA, Alobaid TA, et al. (2024) The nutritional status of mycetoma affected patients seen at the Mycetoma Research Center, Sudan. PLoS Negl Trop Dis 18(1): e0011726. https://doi.org/10.1371/journal.pntd.0011726

Editor: Joshua Nosanchuk, Albert Einstein College of Medicine, UNITED STATES

Received: October 18, 2023

Accepted: December 18, 2023

Published: January 2, 2024

**Peer Review History:** PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0011726

**Copyright:** © 2024 Gabani et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** "All data are in the manuscript and Supporting information files".

**Funding:** The author(s) received no specific funding for this work.

**RESEARCH ARTICLE** 

# The nutritional status of mycetoma affected patients seen at the Mycetoma Research Center, Sudan

Manal Hassan Gabani<sup>1</sup>, Arwa Abdelraouf Ahmed<sup>1</sup>, Alshaima Abdelelah Hassan<sup>1</sup>, Mona Abdelrahim Abdalla<sup>1</sup>, Samar Abdelmahmoud Mustafa<sup>1</sup>, Tasneem Abdelmutalab Alobaid<sup>1</sup>, Abrar Adam Khatir<sup>1</sup>, Reell Mukhlis Mohammed<sup>1</sup>, Nehal Ibrahim Awad<sup>1</sup>, Tanzeel Alqurashi Abdellateef<sup>1</sup>, Abeer Hassan<sup>1</sup>, Eiman Siddig Ahmed<sup>1</sup>, Mohammed Zain Ali<sup>2</sup>, Ahmed Hassan Fahal<sup>1</sup>,<sup>1</sup>\*

1 Mycetoma Research Center, University of Khartoum, Khartoum, Sudan, 2 The School of Health Sciences, Ahfad University for Women, Omdurman, Sudan

\* ahfahal@mycetoma.edu.sd, ahfahal@gmail.com

# Abstract

Nutrition plays a critical and crucial role in addressing neglected tropical diseases (NTDs) and their complications, as they often contribute to malnutrition, which can worsen the impact of these conditions. Therefore, it is necessary to investigate the nutritional status of mycetoma patients, which has not been explored previously.

This descriptive cross-sectional hospital-based study was conducted at the Mycetoma Research Center (MRC), University of Khartoum, Sudan. The study included 179 confirmed mycetoma patients and an equal number of age- and sex-matched normal controls. The nutritional status of the mycetoma patients was assessed and compared with that of the control group. The majority of the patients were young adults with varying educational levels, predominantly from Central Sudan. The foot was the most commonly affected part; most patients had lesions more than 10 cm in diameter.

The Body Mass Index (BMI) was calculated for both study groups, revealing that 43.5% of the patients and 53.6% of controls had a normal BMI. Furthermore, 36% of patients were underweight, contrasting with only 11% in the control group. Correlation analyses indicated no significant associations between BMI and age groups, educational levels, daily meals, food quantity, and appetite in the study population (p > 0.05). Similarly, no significant differences were observed in BMI concerning disease duration and affected sites (p = 0.0577). The Kruskal-Wallis test did not reveal significant differences in BMI means among the groups.

The study revealed that most participants consumed three meals daily, and the control group showed a more robust appetite and consumed more food than the patient group (p = 0.005). Nevertheless, there were no significant differences in the consumption of different food types between the patient and control groups and among different BMI categories (p = 0.025 and 0.040, respectively).

**Competing interests:** The authors have declared that no competing interests exist.

#### Author summary

Mycetoma primarily affects low-income people in limited-resource communities in tropical and subtropical regions. The affected mycetoma are of low socioeconomic and health education status, and their localities' health and medical facilities are frequently inadequate. Hence, they present with massive, complicated disease; thus, the treatment outcome is suboptimal. Mycetoma has many disabilities and psychosocial effects, which may affect the patient nutritional status. Furthermore, most of the mycetoma epidemiological risk factors are indistinct and unclear. With this background, this study was conducted to determine the effect of the disease on the patients' nutritional status and to determine if nutrition has a role in mycetoma susceptibility. The study included 179 patients with confirmed mycetoma and age and sex-matched 179 controls from their communities. The mean patients' height was 160±14.31cm, and for the control was 166.78±10.97 cm. The patients' body weight mean was 56.09±17.67kg; for the control, it was 67.53±16.41kg, and no statistically significant differences existed. The body mass index showed a higher percentage of underweight individuals in the patients group, which was statistically significant. No significant correlations existed between the study population BMI groups and their demographic characteristics. There were many similarities between the studied patients and the control regarding the dietary habits and diet intake. More in-depth studies are needed to determine the causal and effect relationship between mycetoma and the affected population. Nutritional support and education, while encouraging intake of therapeutic/functional locally available food items, should be fundamental in the different mycetoma management activities and measurements.

## Introduction

Mycetoma is a serious, neglected tropical disease. The leading reason for the neglect is that it affects underprivileged populations in resource-constrained communities of the rural tropical and subtropical regions. It predominately affects individuals of low visibility and voice [1]. Furthermore, in endemic regions, the medical and health facilities are meagre and adequate treatment is lacking or inaccessible [2]. All these lead to a progressive disease with massive morbidities and can be fatal [3]. Mycetoma is a serious, neglected disease that continues to hinder and burden the poor, remote communities [4,5]. Mycetoma is a chronic disabling subcutaneous granulomatous inflammatory disease of fungal and bacterial origin [6]. The inflammatory granuloma progressively spreads to affect the skin, deep tissues and bones, leading to massive tissue damage, destruction and serious morbidities and can be fatal [7,8,9]. The triad of subcutaneous mass, multiple sinuses discharging purulent and sero-purulent discharge frequently containing grains is a disease characteristic [10,11]. The mycetoma occurrence and the risk factors association is still unclear. Likewise, the disease susceptibility, resistance, entry route, and incubation period are unclear [12,13,14].

Mycetoma is a disease of poor communities and remains a public health concern in underdeveloped countries [8,15,16,17]. It is ostracized by society for its destructive nature and its massive negative medical, health and socioeconomic consequences, including disabilityrelated social stigma, poverty and psychosocial disturbances [8,18,19,20].

Presently, the diagnosis of mycetoma is difficult and tedious [21,22]. The available diagnostic tests are invasive, time-consuming, of low sensitivity and specificity, and there is no point of care [23,24,25,26,27,28]. The available treatment of mycetoma is suboptimal, characterised

by a low cure rate and high recurrence and follow-up rates [29,30]. The disease remains with patients for life, and a cure is rarely achieved.

Mycetoma typically affects young adults, the community breadwinners, leading to a vicious cycle of disease, disabilities, and poverty, and may culminate in malnutrition [8,31,32].

It is reported that nutrient supplementations and dietary lifestyle interventions are instrumental in reducing host oxidative stress, strengthening the immune system, mitigating comorbidities and are physiologically beneficial [33]. Nevertheless, in mycetoma, a significant knowledge gap exists concerning the effect of adequate nutrition on disease susceptibility, development, progress, and treatment outcome. With this background, the study set out to batch this gap to provide significant insight into the disease pathogenesis and treatment outcome and consolidate the existing body of literature.

## Patients and methods

#### **Ethics statement**

The study population gave informed written consent, and the study ethical clearance was obtained from the Mycetoma Institutional Review Committee, MRC, No. MRC-IRB, 2021–77.

This descriptive cross-sectional hospital-based study was conducted at the Mycetoma Research Centre, University of Khartoum, Sudan. It included 179 patients with confirmed mycetoma and 179 age and sex-matched normal family members as a control.

The diagnosis of mycetoma was confirmed by meticulous clinical examinations, lesional ultrasound examination, histopathological examination of surgical biopsies and, grains culture and PCR examinations. Confirmed patients and controls were selected with a simple convenience sampling technique. Each patient was paired with control from their families, and age and sex were matched. Institutionalised persons, including prisons, dormitories, nursing homes, and patients with wounds from causes other than mycetoma, were excluded.

An expert clinical nutritionist and eight well-trained medical doctors interviewed the study population. The interview included their demographic characteristics and nutritional habits. They had their weights and heights measured, and their BMI was determined. A pre-designed, validated digital interview-based questionnaire was used for data collection.

Study participants were classified into five groups based on the World Health Organization (WHO) BMI scale classification [34]. The BMI of amputated patients was reported and considered in the data analysis.

The Statistical Packages for Social Sciences (SPSS) programme (version 26, USA) was used to manage the data. Appropriate statistical tests were used for data quantitative analysis, including the paired t-test to compare the nutritional status of the patients and controls. Statistical significance differences (*p*-value, Odds ratio) were calculated.

#### Results

The study included 179 patients with confirmed mycetoma; 105 patients (58.7%) were males and 74 (41.3%) were females. The control group included 136(76%) males, and 43(24%) females. Most patients and controls (55.9%) were young adults in the age group 21–40 years with a mean age of  $34.7\pm17.8$  years and  $36.68\pm14.31$  years, respectively. The study population has different education levels, but most (47.5%) had primary education. Most of the study population was from the Gezira and Khartoum States (51.3%). The studied patients were mostly housewives (26%), students (16.6%), and unemployed (21.5%). In the control group, most of them were students, housewives, 12% each and sellers (9.4%), Table 1. The disease duration ranged from a few months to several years, with a mean of  $8.35\pm8.40$  years.

	Patients		Control		
	No.	%	No.	%	
Gander		·			
Male	105	58.7	136	76.0	
Female	74	41.3	43	24.0	
Age Group (years)		·			
<20	47	26.3	17	9.5	
21-40	78	43.6	100	55.9	
>41	54	30.2	62	34.6	
Education Level		·			
Illiterate	36	20.1	20	11.2	
Quoranic/Khalwa	21	11.7	11	6.1	
Primary-School	85	47.5	44	24.6	
Post-Graduate	27	15.1	54	30.2	
University	08	4.5	41	22.9	
	02	1.1	09	5.0	
Residence	1		1	1	
Blue-Nile	01	0.6%	02	1.1%	
East-Darfur	02	01.1	02	01.1	
Gedarif	03	01.7	04	02.2	
Gezira	52	28.7	58	32.0	
Kassala	03	01.7	05	02.8	
Khartoum	40	22.1	21	11.6	
North-Darfur	03	01.7	05	02.8	
North-Kordofan	17	09.4	17	09.4	
Northern	02	01.1	01	00.6	
Red-Sea	01	0.6	03	01.0	
River-Nile	07	03.9	06	03.3	
Sennar	12	06.6	16	08.8	
South-Darfur	05	02.8	05	02.8	
South Kordofan	00	00.0	01	00.6	
West Darfur	03	01.7	03	01.7	
West Kordofan	01	00.6	01	00.6	
White-Nile	29	16.0	31	17.1	
Occupation		·			
Farmer	21	11.6	14	07.70	
Shepherd	6	03.3	2	01.10	
Worker	8	04.4	18	09.90	
Seller	5	02.8	17	09.40	
Student	30	16.6	22	12.20	
House Wife	47	26.0	22	12.20	
Unemployment	39	21.5	15	08.30	
Other	23	02.7	71	39.20	

#### Table 1. The study population Socio-demographic characteristics.

https://doi.org/10.1371/journal.pntd.0011726.t001

Site	No.	%
Foot	118	65.2
Hand	19	10.5
Knee	11	6.1
Leg	07	3.9
Gluteal	03	1.7
Heal	03	1.7
Hip	02	1.1
Head	02	1.1
Thigh	02	1.1
Testis	01	0.6
Abdominal	01	0.6
Back	01	0.6
Buttock	01	0.6
Foot & Knee	01	0.6
Foot & Leg	01	0.6
Knee & Leg	01	0.6
Rectal	01	0.6
Lesion Size	No.	%
Small < 5cm	19	10.6
Medium 5 -10cm	42	23.5
Massive >10cm	118	65.9

Table 2. The mycetoma site and size distributions.

https://doi.org/10.1371/journal.pntd.0011726.t002

The foot (65.2%), hand (10.5%), knee (6.1%) and leg (3.9%) were affected most in this study. The majority of the studied patients, 118(65.9%), had massive lesions more than 10 cm in diameter, and 42 patients (23.5%) had moderate lesions (5–10 cm in diameter), Table 2.

#### Body mass index

The mean patients' height was  $160\pm14.31$ cm, and for the control, it was  $166.78\pm10.97$ cm. The patients' body weight mean was  $56.09\pm17.67$ kg; for the control, it was  $67.53\pm16.41$ kg. The body mass index for the study population was calculated, revealing that 43.5% of the patients and 53.6% of the control had normal weight. It showed 36% of the patients and 11% of the control were underweight. Table 3

There are no significant correlations between the study population BMI groups and their age groups, educational level, meals taken per day, amount of food taken and their appetite, p>0.05 and the disease duration and sites, p = 0.0577. Kruskal Wallis test revealed no significant difference between the means of BMI groups. (Results shown in S1).

Table 3.	The distribution	of the study p	population a	according to	the BMI.
----------	------------------	----------------	--------------	--------------	----------

Group	BMI (kg/m2)	Patients No. (%)	Control No. (%)	
Underweight	<18.50	64 (35.7%)	19 (11%)	
Normal Weight	18.5–24.9	78(43.5%)	96 (53.6%)	
Overweight	25.0-29.9	25 (13.9%)	46 (25.6%)	
Obese I	30.0-34.9	08 (4.4%)	14 (7.1%)	
Obese II	≥35.0	04 (2.2%)	04 (2.2%)	

https://doi.org/10.1371/journal.pntd.0011726.t003

#### The dietary intake

The study showed most of the study population took three meals per day. The control group had a better appetite and ate more food than the patients group, p = 0.005. The study showed that 38% of the patients and 48% of the control group took meat daily. Regarding cheese intake, most of the study population did not take it daily, and 30% of the patients and control took rice 2 to 3 times per week. Less than 20% of patients and control did not take lemon daily. Less than 20% of patients took fava-beans daily, whereas more than 20% of control did, Table 4.

The dietary intake and habits distribution according to the BMI classification are shown in (Results shown in S1).

The study revealed no significant differences in cereal consumption between patients and the control group, as well as among different BMI groups (p = 0.025 and 0.040, respectively). Sunflower oil intake was significantly associated with the BMI of the control group (p = 0.025), while meat and eggs showed significant associations with the control group's BMI (p = 0.044and 0.040, respectively). No significant association was observed between dairy product consumption and population BMI. Pigeon peas in the legumes group correlated significantly with both patients and control groups, as well as BMI groups (p = 0.004 and 0.025, respectively). Fruits showed insignificant correlations with patients and control. In the drinks and juices group, bottled drinks significant associations with both control group (p = 0.015). Cooked vegetables had significant associations with both control and patient groups and the BMI (p = 0.005 and 0.018, respectively).

There were no significant differences in the consumption of various cereal types between the patients and control groups, as well as among different BMI groups, p = 0.025 and 0.040, respectively. Regarding oil-rich foods, a higher intake of sunflower oil was significantly associated with the BMI of the control group (p = 0.025). Among animal products, only meat and eggs showed significant associations with the BMI of the control group (p = 0.044 and 0.040, respectively), while there was no significant association between dairy product consumption and population BMI. In the legumes group, the consumption of pigeon peas significantly correlated with both patients and control groups, as well as BMI groups, p = 0.004 and 0.025, respectively. All types of fruits consumed in the study showed insignificant correlations with patients and control BMIs. In the drinks and juices food group, the consumption of bottled drinks significantly correlated with the BMI of the control group (p = 0.015). The consumption of cooked vegetables had significant associations with both control and patients groups, as well as BMI (p = 0.005 and 0.018, respectively).

Item	Ranges	C N	Control No. (%)		Patients No. (%)	
Meals taken per day	1	02	01.1%	00	00.0%	
	2	70	39.1%	79	44.2%	
	3	107	59.7%	100	55.8%	
Food amount taken	Less	27	15.0%	44	24.3%	
	Same	121	67.5%	123	68.0%	
	More	31	17.3%	14	07.7%	
Appetite	Low	26	14.5%	56	31.2%	
	Moderate	120	67.0%	109	60.8%	
	High	33	18.4%	14	07.8%	

#### Table 4. The study population's dietary habits.

https://doi.org/10.1371/journal.pntd.0011726.t004

### Discussion

Several NTDs, such as intestinal helminthiasis and schistosomiasis, frequently induce serious nutrients malabsorption and malnutrition [35]. These infections can lead to malabsorption of many essential vitamins, minerals and other elements from the diet [36]. Furthermore, malnutrition can increase the susceptibility to many NTDs and other infections. Thus, adequate nutrition is indispensable for maintaining dynamic immune responses to defend the human body against such infections [37]. Furthermore, proper nutrition is vital to support the affected patients in dealing with the NTDs and minimise the long-term NTDs complications, and mycetoma is not an exception.

In this study, 35.4% of the studied patients were underweight, 43.5% had normal weight, 13.8% were overweight, and 7.8% were obese. The incidence of being underweight is higher compared to the control (10.5%). However, it is difficult to determine the causal and effect relationship in this group. The low weight can be disease-induced, or it can be a disease risk factor, and this demands more investigations. The mycetoma patients' weight distributions reported in this study differ from the weight of those with leprosy, a mycetoma similar disease. One report from Brazil showed that 60% of leprosy patients had a normal BMI, 14% were underweight, and 26% were overweight or obese [38]. The cause is unclear, but the studied populations probably lacked homogeneity. However, maintaining a healthy body weight is vital, as excessive weight can strain the mycetoma-affected areas and potentially worsen deformities and disabilities.

This study showed no clear association between the patients' BMI and mycetoma. The literature revealed contraindicating reports on the effects of nutrition on other NTDs. In leprosy, many studies reported normal BMI in most patients, and in some studies, a high incidence of obesity (60%) was reported [39,40]. These studies showed no association between leprosy reactional states and BMI. Furthermore, BMI failed to correlate with the presence of reaction leprosy and deformities or multibacillary disease in the reported studies (p>0.05) [41,42]. Another study from Indonesia reported no significant association between poor nutrition and leprosy (p = 0.008, OR = 0.422) [43]. However, a substantial association between undernutrition (BMI<18.5) and leprosy and WHO grade 2 disability was reported [44]. One study reported a higher incidence of growth disorders and anaemia in children with tuberculous meningitis complicated with pulmonary and abdominal tuberculosis. In these patients, the nutritional status was one of the causes of abandoning treatment [45].

In many NTDs, food and its availability are a substantial prevention and management influence [46]. Most of the NTDs affected patients from low socioeconomic strata have lifelong food insecurity, with enormous psychosocial and physical disabilities to access the available meagre food [47]. It is clear nutrition influences the patients' immunity, but the mechanism is unclear. Furthermore, the exact role of nutrition on susceptibility and development of mycetoma and other NTDs is still indistinct and complicated.

In leprosy, case-control studies on socioeconomic factors showed that the disease is sometimes associated with food shortage [48,49]. Inadequate food intake leads to reduced intake of carbohydrates, proteins, fats, vitamins, and minerals, and this nutritional deficiency impairs the immune system against several infections [50]. However, in mycetoma, that is unclear. As reported in Brazil, the socioeconomic policies and population health conditions in developing nations are closely related and can influence the NTDs new case detection. The high coverage of the Brazilian Conditional Cash Transfer and Primary Health Care programmes significantly reduced the leprosy new case detection; however, the underlying mechanism between food, poverty, and leprosy remains unclear [51]. The various diet types contain different nutritional elements such as protein, fat, carbohydrates, vitamins, minerals, trace elements, phytochemicals, probiotics and antioxidants. All have different roles in supporting body functions during stressful events, particularly chronic infections such as mycetoma. [52]. This study documented that the food type, the study population takes is the normal Sudanese diet, and more in-depth studies are needed to quantify that and see if some nutritional elements are deficient, if any, and their effects on the disease progress and treatment outcomes.

Nutrient supplementation has been influential in suppressing the host oxidative stress, strengthening immune responsiveness and mitigating potential adverse events in leprosy and diabetes [53]. Furthermore, the implications of vitamin and mineral supplementation in leprosy have demonstrated a significant reduction in symptoms and therapeutic requirements [54,55,56]. As mycetoma shares many characteristics of leprosy, it is worth trying nutrient supplementations for these patients to improve the disease progress and treatment outcome.

In this study, only clinical assessment was used to determine the nutritional status of the studied mycetoma patients, which is a limitation. Further studies, including biochemical nutritional parameters and body composition by bioelectrical impedance analysis, are needed for better insight into the effects of patients' nutritional status on the disease's clinical progress and treatment outcome. The successful intake of probiotics as ingestible adjuvants and immune modulators for antiviral immunity and management of SARS-CoV-2 infection and COVID-19 is worth studying in mycetoma.[57].

In conclusion, nutritional assessment and education should be integral parts of the mycetoma management guidelines. Patients should have optimal nutritional status, which is essential for appropriately functioning immune response, adequate response to mycetoma chronic infection and postoperative wound healing. Further in-depth research work is needed to determine the nutritional elements' effects on mycetoma pathogenesis and the treatment outcome.

# **Supporting information**

S1 Table. The dietary intake and habits of the study population according to the BMI classification.

(DOCX)

**S2** Table. Food items consumed by the study population by Body Mass Index classification. (DOCX)

# **Author Contributions**

- **Conceptualization:** Manal Hassan Gabani, Arwa Abdelraouf Ahmed, Alshaima Abdelelah Hassan, Mona Abdelrahim Abdalla, Samar Abdelmahmoud Mustafa, Tasneem Abdelmutalab Alobaid, Abrar Adam Khatir, Reell Mukhlis Mohammed, Nehal Ibrahim Awad, Tanzeel Alqurashi Abdellateef, Abeer Hassan, Eiman Siddig Ahmed, Mohammed Zain Ali, Ahmed Hassan Fahal.
- **Data curation:** Manal Hassan Gabani, Arwa Abdelraouf Ahmed, Alshaima Abdelelah Hassan, Mona Abdelrahim Abdalla, Samar Abdelmahmoud Mustafa, Tasneem Abdelmutalab Alobaid, Abrar Adam Khatir, Reell Mukhlis Mohammed, Nehal Ibrahim Awad, Tanzeel Alqurashi Abdellateef, Abeer Hassan, Eiman Siddig Ahmed, Mohammed Zain Ali, Ahmed Hassan Fahal.

Formal analysis: Manal Hassan Gabani, Arwa Abdelraouf Ahmed, Alshaima Abdelelah Hassan, Mona Abdelrahim Abdalla, Samar Abdelmahmoud Mustafa, Tasneem Abdelmutalab

Alobaid, Abrar Adam Khatir, Reell Mukhlis Mohammed, Nehal Ibrahim Awad, Tanzeel Alqurashi Abdellateef, Abeer Hassan, Eiman Siddig Ahmed, Mohammed Zain Ali, Ahmed Hassan Fahal.

- **Investigation:** Manal Hassan Gabani, Arwa Abdelraouf Ahmed, Alshaima Abdelelah Hassan, Samar Abdelmahmoud Mustafa, Tasneem Abdelmutalab Alobaid, Abrar Adam Khatir, Reell Mukhlis Mohammed, Nehal Ibrahim Awad, Tanzeel Alqurashi Abdellateef, Abeer Hassan, Eiman Siddig Ahmed, Mohammed Zain Ali, Ahmed Hassan Fahal.
- **Methodology:** Manal Hassan Gabani, Arwa Abdelraouf Ahmed, Alshaima Abdelelah Hassan, Mona Abdelrahim Abdalla, Samar Abdelmahmoud Mustafa, Tasneem Abdelmutalab Alobaid, Abrar Adam Khatir, Reell Mukhlis Mohammed, Nehal Ibrahim Awad, Tanzeel Alqurashi Abdellateef, Abeer Hassan, Eiman Siddig Ahmed, Mohammed Zain Ali, Ahmed Hassan Fahal.

Project administration: Manal Hassan Gabani, Mohammed Zain Ali.

Supervision: Eiman Siddig Ahmed, Mohammed Zain Ali, Ahmed Hassan Fahal.

Validation: Mohammed Zain Ali.

- Visualization: Manal Hassan Gabani, Arwa Abdelraouf Ahmed, Alshaima Abdelelah Hassan, Mona Abdelrahim Abdalla, Samar Abdelmahmoud Mustafa, Tasneem Abdelmutalab Alobaid, Abrar Adam Khatir, Reell Mukhlis Mohammed, Nehal Ibrahim Awad, Tanzeel Alqurashi Abdellateef, Abeer Hassan, Eiman Siddig Ahmed, Mohammed Zain Ali, Ahmed Hassan Fahal.
- Writing original draft: Manal Hassan Gabani, Arwa Abdelraouf Ahmed, Alshaima Abdelelah Hassan, Mona Abdelrahim Abdalla, Samar Abdelmahmoud Mustafa, Tasneem Abdelmutalab Alobaid, Abrar Adam Khatir, Reell Mukhlis Mohammed, Nehal Ibrahim Awad, Tanzeel Alqurashi Abdellateef, Abeer Hassan, Eiman Siddig Ahmed, Mohammed Zain Ali, Ahmed Hassan Fahal.
- Writing review & editing: Manal Hassan Gabani, Arwa Abdelraouf Ahmed, Alshaima Abdelelah Hassan, Mona Abdelrahim Abdalla, Samar Abdelmahmoud Mustafa, Tasneem Abdelmutalab Alobaid, Abrar Adam Khatir, Reell Mukhlis Mohammed, Nehal Ibrahim Awad, Tanzeel Alqurashi Abdellateef, Abeer Hassan, Eiman Siddig Ahmed, Mohammed Zain Ali, Ahmed Hassan Fahal.

#### References

- 1. Fahal AH. Mycetoma. In Mora-Montes HM, Leila M. Lopes-Bezerra L (Editors) Current Progress in Medical Mycology, Springer Nature 2017 pp 355–380.
- Zijlstra EE, van de Sande WWJ, Welsh O, Mahgoub ES, Goodfellow M, Fahal AH. Mycetoma: a unique neglected tropical disease. Lancet Infect Dis. 2016; 16(1):100–112. <u>https://doi.org/10.1016/S1473-</u> 3099(15)00359-X PMID: 26738840
- Fahal AH, Hassan MA. Mycetoma. Br J Surg. 1992; 79(11): 1138–1141. <u>https://doi.org/10.1002/bjs. 1800791107</u> PMID: 1467883
- 4. Fahal AH. Mycetoma thorn on the flesh. Review article. Trans R Soc Trop Med Hyg. 2004; 98(1):3–11.
- 5. Fahal AH. Mycetoma. In Richard, Guerrant, Walker, Peter, Tropical Infectious Diseases: Principles, Pathogens and Practice. Third edition, Elsevier Publisher, 2011, 83, 565–568
- 6. Fahal AH. Mycetoma. Review article. Khartoum Med J. 2011; 4(1): 514–523.
- van Belkum A, Fahal A, van de Sande WW. Mycetoma caused by Madurella mycetomatis: a completely neglected medico-social dilemma. Adv Exp Med Biol. 2013; 764:179–89. https://doi.org/10.1007/978-1-4614-4726-9\_15 PMID: 23654067

- Abbas M, Scolding PS, Yosif AA, EL Rahman RF, EL-Amin MO, Elbashir MK, et al. The disabling consequences of Mycetoma. PLoS Negl Trop Dis 2018, 12(12): e0007019. https://doi.org/10.1371/journal. pntd.0007019 PMID: 30532253
- Ahmed AO, van Leeuwen W, Fahal A, van de Sande W, Verbrugh H, van Belkum A. Mycetoma caused by Madurella mycetomatis: a neglected infectious burden. Lancet Infect Dis. 2004; 4(9):566–74. <a href="https://doi.org/10.1016/S1473-3099(04)01131-4">https://doi.org/10.1016/S1473-3099(04)01131-4</a> PMID: 15336224
- Fahal AH, Suliman SH, Hay R. Mycetoma: The Spectrum of Clinical Presentation. Trop Med Infect Dis. 2018, 4; 3(3):97. https://doi.org/10.3390/tropicalmed3030097 PMID: 30274493
- Fahal AH. Mycetoma. In Williams NS, Bulstrode C, O'Connell PR. Bailey and Love's Short Practice of Surgery 26E. 26th ed. Taylor & Francis Ltd.; London, UK: 2013, pp. 84–88.
- Fahal AH, van de Sande WWJ. The Epidemiology of Mycetoma. Curr Fungal Infect Rep. 2012; 6:320– 326.
- Fahal AH, Shaheen S. Jones DH. Orthopaedic Aspects of Mycetoma: Instructional Review. Bone Joint J 2014; 96-B: 420–5.
- Nenoff P, van de Sande WW, Fahal AH, Reinel D, Schöfer H. Eumycetoma and actinomycetoma—an update on causative agents, epidemiology, pathogenesis, diagnostics and therapy. J Eur Acad Dermatol Venereol. 2015; 29(10):1873–83. https://doi.org/10.1111/jdv.13008 PMID: 25726758
- Bonifaz A, Tirado-Sánchez A, Calderón L, Saúl A, Araiza J, Hernández M, et al. Mycetoma: Experience of 482 Cases in a Single Center in Mexico. PLoS Negl Trop Dis. 2014, 8(8): e3102. <u>https://doi.org/10. 1371/journal.pntd.0003102</u> PMID: 25144462
- Omer RF, Seif EL Din N, Abdel Rahim FA, Fahal AH. Hand Mycetoma: The Mycetoma Research Centre Experience and Literature Review. PLoS Negl Trop Dis. 2016; 10(8): e0004886. <u>https://doi.org/10. 1371/journal.pntd.0004886</u> PMID: 27483367
- 17. Fahal AH. Mycetoma: Clinico-Pathological Monograph. University of Khartoum Press; Khartoum, Sudan: 2006. pp. 40–49.
- Hassan MA, Fahal AH. Mycetoma. In Tropical Surgery. Kamil R, Lumbly J, editors. Westminster Publication Ltd.; London, UK: 2004. pp. 30–45.
- López-Martínez R, Méndez-Tovar LJ, Bonifaz A, Arenas R, Mayorga J, Welsh O, et al. Update on the epidemiology of mycetoma in Mexico. A review of 3933 cases. Gac Med Mex. 2013; 149(5):586–92.
- Fahal AH, Sharfy ARA. Vulval mycetoma: a rare cause of bladder outlet obstruction. Trans R Soc Trop Med Hyg. 1998; 92: 652–653. https://doi.org/10.1016/s0035-9203(98)90798-2 PMID: 10326112
- Ahmed AA, van de Sande W, Fahal AH. Mycetoma laboratory diagnosis: Review article. PLoS Negl Trop Dis. 2017; 11(8): e0005638. <u>https://doi.org/10.1371/journal.pntd.0005638</u> PMID: 28837657
- 22. EL Hag IA, Fahal AH, Gasim ET. Fine needle aspiration cytology of mycetoma. Acta Cytol. 1996 May-Jun; 40(3):461–4. https://doi.org/10.1159/000333899 PMID: 8669179
- Lim W, Siddig E, Eadie K, Nyuykonge B, Ahmed S, Fahal A, et al. The development of a novel diagnostic PCR for Madurella mycetomatis using a comparative genome approach. PLoS Negl Trop Dis. 2020 Dec 16; 14(12):e0008897 https://doi.org/10.1371/journal.pntd.0008897 PMID: 33326425
- Siddig EE, Verbon A, Bakhiet S, Fahal AH, van de Sande WWJ. The developed molecular biological identification tools for mycetoma causative agents: An update. Acta Trop. 2022 Jan; 225:106205. https://doi.org/10.1016/i.actatropica.2021.106205 PMID: 34687643
- Abd El-Bagi ME, Fahal AH. Mycetoma revisited. Incidence of various radiographic signs. Saudi Med J. 2009; 30(4):529–33. PMID: 19370281.
- EL Shamy ME, Fahal AH, Shakir MY, Homedia MMA. New MRI Grading System for the Diagnosis and Management of Mycetoma. Trans R Soc Trop Med Hyg. 2012; 106(12):738–42. https://doi.org/10. 1016/j.trstmh.2012.08.009 PMID: 22981317
- Fahal AH, Sheik HE, Homeida MM, Arabi YE, Mahgoub ES. Ultrasonographic imaging of mycetoma. Br J Surg. 1997 Aug; 84(8):1120–2. PMID: 9278658
- van de Sande WW, Fahal AH, Goodfellow M, Mahgoub el S, Welsh O, Zijlstra EE. Merits and pitfalls of currently used diagnostic tools in mycetoma. PLoS Negl Trop Dis. 2014 Jul 3; 8(7):e2918. <u>https://doi.org/10.1371/journal.pntd.0002918 PMID: 24992636</u>.
- Zein HAM, Fahal AH, Mahgoub ES, EL Hassan T, Abdel Rahman ME, The Predictors of Cure, Amputation & Follow-up dropout among Mycetoma Patients as seen at The Mycetoma Research Centre, University of Khartoum. Trans R Soc Trop Med Hyg. 2012; 106(11):639–44.
- Elkheir LYM, Haroun R, Mohamed MA, Fahal AH. Madurella mycetomatis causing eumycetoma medical treatment: The challenges and prospects. PLoS Negl Trop Dis. 2020; 14(8): e0008307. https://doi. org/10.1371/journal.pntd.0008307 PMID: 32853199

- Fahal AH. Mycetoma: A global medical and socioeconomic dilemma. PLoS Negl Trop Dis. 2017; 11(4): e0005509. https://doi.org/10.1371/journal.pntd.0005509 PMID: 28426654
- Bakhiet SM, Fahal AH, Musa AM, Mohamed ESW, Omer RF, Ahmed ES, et al. A holistic approach to the mycetoma management. PLoS Negl Trop Dis. 2018 May 10; 12(5):e0006391. https://doi.org/10. 1371/journal.pntd.0006391 eCollection 2018 May. PMID: 29746460
- **33.** Klowak M, Boggild AK. A review of nutrition in neuropathic pain of leprosy. Therapeutic Advances in Infectious Disease. 2022; 9. https://doi.org/10.1177/20499361221102663 PMID: 35677111
- 34. https://www.who.int/europe/news-room/fact-sheets/item/a-healthy-lifestyle—who-recommendations
- Degarege A, Erko B, Negash Y, Animut A. Intestinal Helminth Infection, Anemia, Undernutrition and Academic Performance among School Children in Northwestern Ethiopia. Microorganisms. 2022 Jul 5; 10(7):1353. https://doi.org/10.3390/microorganisms10071353 PMID: 35889072.
- Crompton DW, Whitehead RR. Hookworm infections and human iron metabolism. Parasitology. 1993; 107:S137–S145. https://doi.org/10.1017/s0031182000075569 PMID: 8115178
- Foolchand A, Ghazi T, Chuturgoon AA. Malnutrition and dietary habits alter the immune system which may consequently influence SARS-CoV-2 virulence: A Review. Int J Mol Sci. 2022; 28; 23(5):2654. https://doi.org/10.3390/jims23052654 PMID: 35269795.
- Montenegro RM, Zandonade E, Molina Mdel C, Diniz LM: Reactional state and nutritional profile among leprosy patients in the primary health care system, Greater Vitória, Espírito Santo State, Brazil. Cad Saude Publica. 2012, 28:31–8. https://doi.org/10.1590/s0102-311x2012000100004 PMID: 22267063
- Teixeira CS, Medeiros DS, Alencar CH, Ramos Júnior AN, Heukelbach J. Nutritional aspects of people affected by leprosy, between 2001 and 2014, in semi-arid Brazilian municipalities. Cien Saude Colet. 2019, 24:2431–41. https://doi.org/10.1590/1413-81232018247.19642017 PMID: 31340262
- Rao PS, John AS: Nutritional status of leprosy patients in India. Indian J Lepr. 2012, 84:17–22. PMID: 23077779
- Jindal R, Nagrani P, Chauhan P, Bisht YS, Sethi S, Roy S. Nutritional Status of Patients With Leprosy Attending a Tertiary Care Institute in North India. Cureus. 2022 Mar 16; 14(3):e23217. https://doi.org/ 10.7759/cureus.23217 PMID: 35449687.
- 42. Dwivedi VP, Banerjee A, Das I, Saha A, Dutta M, Bhardwaj B, et al. Diet and nutrition: An important risk factor in leprosy. Microb Pathog. 2019; 137:103714. https://doi.org/10.1016/j.micpath.2019.103714 Epub 2019 Sep 4. PMID: 31493502.
- 43. Listiawan MY, Endaryanto A, Prakoeswa CRS. Correlation Analysis between Household Hygiene and Sanitation and Nutritional Status and Female Leprosy in Gresik Regency. Dermatol Res Pract. 2020 Sep 30; 2020:4379825. https://doi.org/10.1155/2020/4379825 PMID: 33061955.
- Vázquez CMP, Netto RSM, Barbosa KBF, de Moura TR, de Almeida RP, Duthie MS, et al. Micronutrientes que influyen en la respuesta immune en la lepra. Nutr Hosp 2014; 29: 26–36.
- Gao Z, Liu Q, Deng Q, Kong L, Liu Y. Growth and anemia among children with tuberculosis infection at different sites in Southwest China. Front. Pediatr. 2023; 11:1188704. <u>https://doi.org/10.3389/fped.</u> 2023.1188704 PMID: 37397155
- Hall A, Zhang Y, Macarthur C, Baker S. The role of nutrition in integrated programs to control neglected tropical diseases. BMC Med. 2012; 25;10:41. https://doi.org/10.1186/1741-7015-10-41 PMID: 22533927.
- Ochola EA, Karanja DMS, Elliott SJ. The impact of Neglected Tropical Diseases (NTDs) on health and wellbeing in sub-Saharan Africa (SSA): A case study of Kenya. PLoS Negl Trop Dis. 2021 Feb 11; 15 (2):e0009131. https://doi.org/10.1371/journal.pntd.0009131 PMID: 33571200.
- Wagenaar I, van Muiden L, Alam K, Bowers R, Hossain MA, Kispotta K, et al. Diet-related risk factors for leprosy: a case-control study. PLoS Negl Trop Dis. 2015 May 12; 9(5):e0003766. https://doi.org/10. 1371/journal.pntd.0003766 PMID: 25965879.
- Kerr-Pontes LRS, Barreto ML, Evangelista CMN, Rodrigues LC, Heukelbach J, Feldmeier H. Socioeconomic, environmental, and behavioural risk factors for leprosy in North-east Brazil: results of a case– control study. Int. J. Epidemiol. 2006; 34; 994–1000. <u>https://doi.org/10.1093/ije/dyl072</u> PMID: 16645029
- Shao T, Verma HK, Pande B, Costanzo V, Ye W, Cai Y, et al. Physical Activity and Nutritional Influence on Immune Function: An Important Strategy to Improve Immunity and Health Status. Front Physiol. 2021 Oct 8; 12:751374. https://doi.org/10.3389/fphys.2021.751374 PMID: 34690818.
- Nery JS, Pereira SM, Rasella D, Penna MLF, Aquino R, Rodrigues LC, et al. Effect of the Brazilian Conditional Cash Transfer and Primary Health Care Programs on the New Case Detection Rate of Leprosy. PLoS Negl Trop Dis. 2014; 8(11): e3357. https://doi.org/10.1371/journal.pntd.0003357 PMID: 25412418
- Pecora F, Persico F, Argentiero A, Neglia C, Esposito S. The Role of Micronutrients in Support of the Immune Response against Viral Infections. Nutrients. 2020 Oct 20; 12(10):3198. https://doi.org/10. 3390/nu12103198 PMID: 33092041

- 53. da Silva PHL, de Castro KKG, Mendes MA, Calvo TL, Leal JMP, Hacker MAV, et al. Increased oxidative stress in elderly leprosy patients is related to age but not to bacillary load. PLoS Negl Trop Dis. 2021 Mar 9; 15(3):e0009214. https://doi.org/10.1371/journal.pntd.0009214 PMID: 33690671.
- 54. Klowak M, Boggild AK. A review of nutrition in neuropathic pain of leprosy. Ther Adv Infect Dis. 2022 Jun 1; 9:20499361221102663. https://doi.org/10.1177/20499361221102663 PMID: 35677111.
- 55. Vijayaraghavan R, Suribabu CS, Sekar B, Oommen PK, Kavithalakshmi SN, Madhusudhanan N, et al. Protective role of vitamin E on the oxidative stress in Hansen's disease (Leprosy) patients. Eur J Clin Nutr. 2005 Oct; 59(10):1121–8. https://doi.org/10.1038/sj.ejcn.1602221 PMID: 16015260.
- 56. Haase H, Overbeck S and Rink L. Zinc supplementation for the treatment or prevention of disease: current status and future perspectives. Exp Gerontol 2008; 43: 394–408. https://doi.org/10.1016/j.exger. 2007.12.002 PMID: 18221847
- 57. Tomkinson S, Triscott C, Schenk E, Foey A. The Potential of Probiotics as Ingestible Adjuvants and Immune Modulators for Antiviral Immunity and Management of SARS-CoV-2 Infection and COVID-19. Pathogens 2023, 12(7), 928. https://doi.org/10.3390/pathogens12070928 PMID: 37513775